 Vc/23-Aug-04
Week35 starting 23Aug  04

SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday 23 aug04  ~15.50 hrs 

	TBA1
	MBAL3199
	Prep 0 Scan Bench
	299.1

	TBA2
	MBBL3102
	PT 5.1 1.5 KA Quench LF
	1.9

	TBB1
	SSS018
	PT 3 QH Measure
	1.9

	TBB2
	MBAL3186
	PT 12 Warm Up
	210.8

	TBC1
	MBBL2104
	PT 12 Warm Up
	84.7

	TBC2
	MBBL1143
	Prep 5 Pumping + Cool Down
	296.3

	TBD1
	SSS044
	Anticryostat Alignment
	298.6

	TBD2
	SSS012
	ICS 5 Traveller Last
	298

	TBE1
	MBAL3201
	ICS 3 Traveller after warm test
	300.3

	TBE2
	MBBL3197
	Prep 5 Pumping + Cool Down
	36.8

	TBF1
	MBAL2057
	Prep 5 Pumping + Cool Down
	298.5

	TBF2
	MBBL2084
	PT 10 Min. NRJ 11850 A
	14.6


8/23/2004 3:51:21 PM    Temperature data are taken from TBSM18 site
Coordinator this week: Pierre; MTM support on call :  Salem(16-3602).

11 Magnets [ 9  dipoles( 2 with MM)+ 2 SSS]   Disconnected after tests last week ( Sat Midnight to Sat midnight)  
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
· ( SSS work : New issues
· Same Training Rules as for Dipoles from now on- 12 kA rule 
CRYO PRIORITY SETTINGS & PAPER Sheet for OP Crew
Bench status  and tests for now( Monday 23 Aug/7at 15.55 hrs)  & next few days :  see Video Display

	Magnets under Test on 23-08 3:56:52 PM 

	-
	Bench
	Magnet
	Shafts
	Start
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	Last Test
Desc
	
Time
	
I
	
Temp.
	Total

	1
	TBA1
	MBAL3199 
	[image: image1.png]



	23-08 
	1 h.
	-
	-
	-
	-
	ICS 2 Connect Magnet (ICS)
	23-08
(15 : 50) 
	0 A. 
	300 K
	1 hours

	2
	TBA2
	MBBL3102 
	[image: image2.png]



	19-08 
	30 h.
	72 h.
	6 h.
	-
	-
	PT 5.1 1.5 KA Quench LF
	23-08
(14 : 00) 
	1500 A. 
	1.9 K
	108 hours

	3
	TBB1
	SSS018 
	[image: image3.png]



	17-08 
	81 h.
	45 h.
	24 h.
	-
	-
	PT 3 QH Measure
	22-08
(17 : 14) 
	0 A. 
	1.9 K
	150 hours

	4
	TBB2
	MBAL3186 
	[image: image4.png]



	18-08 
	30 h.
	49 h.
	50 h.
	3 h.
	-
	PT 12 Warm Up
	23-08
(12 : 38) 
	2 A. 
	Warming
	132 hours

	5
	TBC1
	MBBL2104 
	[image: image5.png]



	18-08 
	27 h.
	74 h.
	19 h.
	5 h.
	-
	PT 12 Warm Up
	23-08
(10 : 22) 
	0 A. 
	Warming
	125 hours

	6
	TBC2
	MBBL1143 
	[image: image6.png]



	21-08 
	41 h.
	-24 h.
	27 h.
	-
	-
	Prep 5 Pumping + Cool Down
	23-08
(12 : 38) 
	0 A. 
	Cooling
	44 hours

	7
	TBD1
	SSS044 
	[image: image7.png]



	18-08 
	118 h.
	-
	-
	-
	-
	Anticryostat Alignment
	23-08
(13 : 09) 
	0 A. 
	300 K
	118 hours

	8
	TBD2
	SSS012 
	[image: image8.png]



	09-08 
	132 h.
	66 h.
	130 h.
	16 h.
	2 h.
	ICS 5 Traveller Last
	23-08
(15 : 29) 
	0 A. 
	300 K
	346 hours

	9
	TBE1
	MBAL3201 
	[image: image9.png]



	23-08 
	8 h.
	-
	-
	-
	-
	ICS 3 Traveller after warm test
	23-08
(15 : 35) 
	0 A. 
	300 K
	8 hours

	10
	TBE2
	MBBL3197 
	[image: image10.png]



	21-08 
	40 h.
	29 h.
	-
	-
	-
	Prep 5 Pumping + Cool Down
	22-08
(10 : 40) 
	0 A. 
	Cooling
	69 hours

	11
	TBF1
	MBAL2057 
	[image: image11.png]



	22-08 
	30 h.
	-17 h.
	20 h.
	-
	-
	Prep 5 Pumping + Cool Down
	23-08
(12 : 37) 
	0 A. 
	Cooling
	33 hours

	12
	TBF2
	MBBL2084 
	[image: image12.png]



	19-08 
	39 h.
	45 h.
	20 h.
	3 h.
	-
	PT 10 Min. NRJ 11850 A
	23-08
(12 : 46) 
	11849 A. 
	1.9 K
	107 hours


Quench Information
 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAL3199
	[image: image13.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBBL3102
	[image: image14.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	SSS018
	[image: image15.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBAL3186
	12327 Amp.
(D1-U)
	12468 Amp.
(D1_U)
	[image: image16.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBBL2104
	11442 Amp.
(D1-L)
	12642 Amp.
(D2-U)
	[image: image17.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBBL1143
	[image: image18.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	SSS044
	[image: image19.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	SSS012
	11592 Amp.
(QI-34)
	12371 Amp.
(QE-34)
	12552 Amp.
(QI-12)
	12626 Amp.
(QE-12)
	12762 Amp.
(QE-12)
	12851 Amp.
(No Data)
	12851 Amp.
([image: image20.jpg]


)
	-
	-
	-
	-
	-
	-

	9
	MBAL3201
	[image: image21.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	MBBL3197
	[image: image22.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBAL2057
	[image: image23.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBBL2084
	11922 Amp.
(D2-U)
	12687 Amp.
(D2-L)
	[image: image24.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-


Results by Week for PREVIOUS week 

	

	Magnets under Test from 15-08-2004 to 21-08-2004 

	-
	Bench
	Magnet
	Shafts
	Start Date .
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	End Date
	Total
	
	

	1
	TBA1
	MBBL3192 
	[image: image25.png]



	11-08-2004 
	45 h.
	31 h.
	16 h.
	30 h.
	2 h.
	16-08-2004 
	120 hours
	
	

	2
	TBA2
	MBBL1107 
	[image: image26.png]



	12-08-2004 
	43 h.
	-29 h.
	114 h.
	22 h.
	11 h.
	19-08-2004 
	168 hours
	
	

	3
	TBB1
	SSS041 
	[image: image27.png]



	02-08-2004 
	132 h.
	32 h.
	95 h.
	17 h.
	73 h.
	17-08-2004 
	360 hours
	
	

	4
	TBB2
	MBBL1121 
	[image: image28.png]



	05-08-2004 
	48 h.
	130 h.
	51 h.
	49 h.
	2 h.
	17-08-2004 
	288 hours
	
	

	5
	TBC1
	MBAL3079 
	[image: image29.png]



	12-08-2004 
	32 h.
	62 h.
	13 h.
	17 h.
	6 h.
	18-08-2004 
	144 hours
	
	

	6
	TBC2
	MBBL2106 
	[image: image30.png]



	10-08-2004 
	43 h.
	36 h.
	9 h.
	14 h.
	8 h.
	15-08-2004 
	120 hours
	
	

	7
	TBC2
	MBBL3198 
	[image: image31.png]



	15-08-2004 
	20 h.
	81 h.
	20 h.
	19 h.
	6 h.
	21-08-2004 
	144 hours
	
	

	8
	TBD1
	SSS007 
	[image: image32.png]



	05-08-2004 
	177 h.
	19 h.
	97 h.
	26 h.
	1 h.
	18-08-2004 
	312 hours
	
	

	9
	TBE1
	MBBL2075 
	[image: image33.png]



	09-08-2004 
	21 h.
	30 h.
	79 h.
	40 h.
	6 h.
	17-08-2004 
	192 hours
	
	

	10
	TBE2
	MBBR1019 
	[image: image34.png]



	14-08-2004 
	33 h.
	61 h.
	30 h.
	15 h.
	8 h.
	20-08-2004 
	144 hours
	
	

	11
	TBF2
	MBBL1131 
	[image: image35.png]



	12-08-2004 
	23 h.
	100 h.
	14 h.
	17 h.
	7 h.
	18-08-2004 
	144 hours
	
	


Quench Information
	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBL3192
	12497 A.
(D2-U)
	12850 A.
[image: image36.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBBL1107
	11679 A.
(D1_U)
	12029 A.
(D2-U)
	12346 A.
(D1-L)
	[image: image37.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	SSS041
	12281 A.
(QI-12)
	12610 A.
(QE-12)
	12446 A.
(QE_12)
	12850 A.
(No Data)
	[image: image38.png]



	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBBL1121
	11955 A.
(D1-U-S06)
	12193 A.
(D1_U_S06)
	12592 A.
(No Data)
	12539 A.
(D1-U-S06)
	12720 A.
(D2-L-S09)
	[image: image39.png]



	-
	-
	-
	-
	-
	-
	-

	5
	MBAL3079
	12832 A.
(D2-U)
	12657.2 A.
(D2-L)
	[image: image40.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBBL2106
	12139 A.
(D2_L)
	12816 A.
(D1_U)
	[image: image41.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	MBBL3198
	12636 A.
(D1-L)
	12850 A.
[image: image42.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	SSS007
	12429 A.
(QI_12-34)
	12850 A.
[image: image43.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBBL2075
	10852 A.
(D1_U_S01)
	12290 A.
(D2-L-S01)
	[image: image44.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	MBBR1019
	12483 A.
(D2-U)
	12245 A.
(D2_L)
	[image: image45.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBBL1131
	12026 A.
(D1-L)
	12301 A.
(D1-L-MRB)
	[image: image46.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-


General Problems last week :  
	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday_17_A
	TBF2
	1131 magnet: LQA setting not activated for the first training quench: quench location lost.
	MTM
	
	

	Tuesday_17_N


	TBD2
	SSS012: the powering of corrector EF could not be done. Power converter goes off when TEMA started. Worked without TEMA then with TEMA again.
	AB/CO
	TEMA?
	Maryline investigated the problem but origin not found.

	Thursday _19_A
	TBF1
	HV test : problem with 1 rack (92): some relays are not activated
	MTM
	
	Rack changed. OK

Richard will have a look Monday 24.

	Thursday _19_A
	TBC2
	Hf DAQ: trigger not received.
	AB-CO
	
	Adrian called and changed a damaged card. 

	Thursday _19_A
	TBF1
	3136 magnet: 1.5 kA quench : no  quench location. The cable of Qloc antenna was connected instead of that of long shaft.
	MTM
	
	

	Thursday _19_N
	TBD2
	SSW at cold on SSS12: strength measurements.

Offset and transfer function wrong.
	MTM
	
	DB File corrupted. DB file replaced by Nicolay.

	Thursday _19_N

Sunday _22_A
	TBB1
	Anti-cryostat alarm before cooling down the magnet. T above 40 C.

The control rack of the anticrysotat did not received signal to warm when the coo-down starts
	MTM
	
	P.Legrand is investigating.

Marlyne called and started locally the warming cycle of anticrysotats.

	Friday_20_M
	TBE1
	PLCSE not OK: QHeater interlock
	MTM
	
	

	Friday_20_N-Saturday_21_N
	TBB2
	Take a long time to cool magnet from 4.2 K to 1.9 K. Origin to be investigated. Y-line not connected?
	ICS?
	
	Cryo piquet:  inform Test ccordinator. Try to complete the test .

	Sunday_22_M

Sunday_22_A
	TBD2
	SSW at cold : Wire broke 2 times during rot wire measurements.
	MTM
	
	In that case : repeat XY measurements and rot wire. It is not needed to repeat the survey.

	Friday_20_M

Sunday _22_M


	TBD2
	SPA at 5000 A after 5 minutes .

Happened two time at 11850 A and 5000 A
	AB-CO
	
	

	Saturday _21_A
	TBA1
	2105 magnet :Diode test at 3 kA: Red alarm : I earth crossing 5 mA 
	MTM
	
	Test coordinator informed: Check the signals and asks for MinNRJ at 4.5 KA. Then OK.

	Sunday _22_A
	TBB1
	Unable to launch hf and lf applications from control room 
	AB-CO
	hf, lf
	Maryline called. She explained that hf and lf application have to be start locally for this bench because of a new version of DAQ (last Tuesday meeting). Alessandro will complete the installation as soon as possible.

	Saturday _21_A
	TBD1
	IAP @ warm: faulty files generated because  DAQ was not put off during changing rotary switch position
	MTM
	
	








